Author's response to reviews

Title: Pneumococcal polyarticular septic arthritis after a single infusion of infliximab in a rheumatoid arthritis patient: a case report

Authors:

Masatoshi Hayashi (hayashimasatoshi@hotmail.com)
Toshihisa Kojima (toshik@med.nagoya-u.ac.jp)
Koji Funahashi (koji2784@sf.starcat.ne.jp)
Daizo Kato (daizo@med.nagoya-u.ac.jp)
Hiroyuki Matsubara (gmyfh191@ybb.ne.jp)
Tomone Shioura (ja-sizukousei-ra@krc.biglobe.ne.jp)
Yasuhide Kanayama (y-kanayama@nifty.com)
Yuji Hirano (yu-hr@sf.commufa.jp)
Masao Deguchi (mg3m-dgc@asahi-net.or.jp)
Toshihisa Kanamono (fwnt3976@mb.infoweb.ne.jp)
Naoki Ishiguro (n-ishi@med.nagoya-u.ac.jp)

Version: 2 Date: 26 December 2011

Author's response to reviews: see over
December 26, 2011

Professor Michael Kidd AM
Editor-in-Chief
Journal of Medical Case Reports

Dear Dr. Michael Kidd AM:

We are grateful to submit our revised manuscript entitled “Pneumococcal polyarticular septic arthritis after a single infusion of infliximab in a rheumatoid arthritis patient: a case report” by Masatoshi Hayashi, Toshihisa Kojima, Koji Funahashi, Daizo Kato, Hiroyuki Matsubara, Tomone Shioura, Yasuhide Kanayama, Yuji Hirano, Masao Deguchi, Toshihisa Kanamono and Naoki Ishiguro to be considered for publication as a case report to Journal of Medical Case Reports.

This revised article includes 12 pages of text and figure legends.

We declare that this manuscript is original, has not been published before and is not currently being considered for publication elsewhere.

We wish to confirm that there are no known conflicts of interest associated with this publication and there has been no significant financial support for this work that could have influence its outcome. I (as Corresponding Author) confirm that the manuscript has been read and approved by all named authors.

We would like to suggest same two peer reviewers (Dr. Rohit Aggarwal and Dr.Mohamed Yassin) you had recommend for our revised manuscript.

We hope you find our manuscript suitable for publication and look forward to hearing from you.

Sincerely,

Masatoshi Hayashi
Department of Orthopedics Surgery and Rheumatology, Nagano Red Cross Hospital, 5-22-1 Wakasato, Nagano, 380-8582, Japan
Telephone: +81-26-226-4131
E-mail address: hayashimatoshi@hotmail.com